

Temi di Discussione

(Working Papers)

Capital destruction, jobless recoveries, and the discipline device role of unemployment

by Marianna Riggi

 $\frac{1}{871}$



Temi di discussione

(Working papers)

Capital destruction, jobless recoveries, and the discipline device role of unemployment

by Marianna Riggi

Number 871 - July 2012

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Silvia Magri, Massimo Sbracia, Luisa Carpinelli, Emanuela Ciapanna, Francesco D'Amuri, Alessandro Notarpietro, Pietro Rizza, Concetta Rondinelli, Tiziano Ropele, Andrea Silvestrini, Giordano Zevi. Editorial Assistants: Roberto Marano, Nicoletta Olivanti.

CAPITAL DESTRUCTION, JOBLESS RECOVERIES, AND THE DISCIPLINE DEVICE ROLE OF UNEMPLOYMENT

by Marianna Riggi^{*}

Abstract

I consider an economy growing along the balanced growth path that is hit by an adverse shock to its capital accumulation process. The model integrates efficiency wages due to imperfect monitoring of the quality of labour in a search and matching framework with methods of dynamic general equilibrium analysis. I show that, depending on the firms' abilities to assess workers' performance, the discipline device role of unemployment may account for sharp declines in employment and jobless recoveries driven by exceptional increases in the work effort of employees. The model also explains why rigid real wages may prevail in equilibrium: the large movements in unemployment are indeed associated with real wage rigidity, which is generated endogenously by efficiency wages.

JEL Classification: E24, E32.

Keywords: jobless recoveries, efficiency wages, productivity, capital depreciation, real wage rigidities.

Contents	
1. Introduction	5
2. Related literature	8
3. The model	
3.1 Preferences	
3.2 The labour market	
3.3 Firms	15
3.4 Aggregate resource constraint	19
4. Capital destruction and unemployment	19
4.1 Unobservable effort	19
4.2 Observable effort	
5. Conclusions	
Figures	
Appendix A	
Appendix B	
Appendix C	
References	

^{*} Bank of Italy, Economic Research and International Relations.

1 Introduction¹

The recession ended officially in June 2009, but the US is still facing an unemployment crisis and the huge human costs of joblessness. Prior to 1990, jobs recovered promptly after the GDP growth turned positive, but the economic downturn of 1990-91 heralded a new era of productivity-led recoveries: the most recent recoveries, following the 1990-91, 2001 and 2007-09 recessions, have failed to create jobs and have been driven by productivity growth (see Figure 1). Particularly rapid productivity growth in the initial quarters after an NBER-dated cyclical trough has been a common feature of almost every postwar recovery.² Peculiar to the last three recoveries was the unusual length of the phase of productivity growth complemented by continuing stalled growth in employment (Gordon, 2010). Labour productivity growth per se is good news but its connection with joblessness in the short run has been highlighted by many economists. For instance, Bernanke (2003) and Yellen (2010) recently conjectured that productivity gains, related also to increased workers' effort, may have contributed to the slowness of the recovery of the labour mar ket^3 and Ohanian (2010a and 2010b) argues that the key to understanding the 2007-2009 recession and its aftermath is developing theories of labour market distortions for explaining why labour input was far below the level consistent with the marginal productivity of labour.

This paper sheds light on the discipline device role of unemployment as a channel that may drive exceptional increases in the work effort of employees in the face of a transi-

³Bernanke (2003) states that: This surprising productivity performance probably reflects both some increase in the long-run rate of productivity growth as well as unmeasured increases in the work effort of employees...Strong productivity growth provides major benefits to the economy in the longer term [...]. But [...] it has also enabled firms to meet the demand for their output without hiring new workers. Thus, in the short run, productivity gains [...] have contributed to the slowness of the recovery of the labor market. Yellen (2010) asserts that [...] the recession has forced businesses to re-examine just about everything they do with an eye toward restraining costs and boosting efficiency. [...] My business contacts describe this as a paradigm shift and they believe it's permanent. This process of implementing new efficiency gains may have only begun and we may be in store for further efficiency improvements and high productivity growth for some time. If so, the rate of job creation will be frustratingly slow.

¹I would like to thank Pierpaolo Benigno, Matteo Bugamelli, Marco Cosconati, Stefano Neri, Francesco Nucci, Alfonso Rosolia, Sergio Santoro, Massimo Sbracia and participants in the CEPR-CREI conference on "Understanding Jobless Recoveries" Universitat Pompeu Fabra, for their helpful suggestions and discussions.

²In this respect, Gordon (1979, 1993 and 2003) identifies two tendencies: the "end-of-expansion effect" and the "early recovery productivity bubble". The former refers to firms' tendency to overhire in the late stages of business expansion, i.e. a tendency for labour input to grow faster than can be explained by output variations. The latter refers to firms' tendency to underhire in the early stages of the recovery, which is associated with a complementary temporary spike in productivity growth.

tory adverse shock to the capital accumulation process,⁴ leading to large and persistent unemployment movements, as the increased efficiency of insiders reduces firms' needs for workers during recoveries.

I consider an economy on a balanced growth path, which is hit by a one-time shock rendering some of its capital stock unproductive. The model incorporates efficiency wages due to imperfect monitoring of workers' performance in a search and matching framework with methods of dynamic general equilibrium analysis. The threat of unemployment induces workers to provide effort. A key difference with respect to Shapiro and Stiglitz's (1984) milestone, where effort is either provided or not provided, is that workers' performance is allowed to vary continuously. Also, while Shapiro and Stiglitz (1984) do not allow for any effect of household wealth on equilibrium wage determination, I assume that the marginal utility of income falls as income rises, thereby allowing for wealth effects and for a balanced growth path along which unemployment is trendless. I show that, in this framework, a transitory shock which causes some fraction of the capital stock to evaporate may lead to large, long-lasting and proportional declines in employment and in output, consumption and investment relative to trend while insiders' performance booms. Jobless recovery may emerge as anemic growth in employment accompanies trend growth in consumption, investment and output.

The model is also able to capture the relative rigidity of real wages observed over the business cycle combined with large movements in unemployment and to explain why rigid real wages may prevail in equilibrium. Ohanian (2010a, 2010b and forthcoming) provides compelling evidence that both in the Great Depression of the 1930s and in the recession of 2008, distortions in the labour market kept wages far above market-clearing levels, resulting in a dramatic decline in employment. Hall (2010) argues that wage unresponsiveness to the large increase in unemployment stands as a major feature of the recent recession. Shimer (2010) points to real wage rigidities as an explanation for jobless recoveries, showing that there is little evidence for a decrease in employment costs relative to trend during the 2001 recession and that detrended real labour costs have been quite stable throughout the recent crisis.⁵ Consistently with these works, the large movements in unemployment in the model economy I propose are synonymous with real wage rigidity, which, however, is generated endogenously by efficiency wages - supporting Solow's (1979) argument - rather than imposed, as instead it is in Shimer (2010).

 $^{^{4}}$ Reasons for looking at the effects of a shock to the capital accumulation process, as in Shimer (2010), are clarified in the next section.

⁵See Figure 2 in Shimer (2010).

The story goes as follows. Efficiency wages imply that workers' performance is strongly countercyclical relative to employment because movements in the quality of work are essentially due to the cyclically varying threat of unemployment.⁶ Suppose that the economy is on its balanced growth path, there is a one-time capital depreciation shock and firms cut back employment. The increased threat of unemployment makes insiders more efficient, inducing them to work harder. This increased efficiency lowers firms' employment needs, resulting in even higher unemployment. This again strengthens the threat and boosts the effort, ultimately leading to amplified unemployment and performance dynamics.

The strength of the discipline device role of unemployment depends on the firm's ability to monitor workers' performance and detect shirkers. If the detection probability is close to zero, the threat of firing as a method of discipline is weak; the mechanism described above is reduced, and effort and employment movements are smoothed. If the detection probability is close to one, the threat of a spell of unemployment is weak, because the surplus accruing to the household from an existing employment relation (which workers would lose were they to be detected shirking and not re-employed) is close to zero and the model tends to become a search and matching economy with workers having zero wage bargaining power and where unemployment plays no role as a method of discipline.⁷ The discipline device role of unemployment is strong if the firm's ability to assess workers' performance is such that the probability of being detected and falling into the pool of the unemployed is considerable and, at the same time, the loss that shirking workers would incur were they to be detected is still extensive. In this case, the mechanism described above is fully at work and the model outcomes in the face of a capital destruction shock resemble a jobless recovery associated with a burst in insiders' performance.

The literature provides much proof of the explosion of a thriving industry devoted to the development of performance appraisal systems in the US after the mid-1980s,⁸ which

⁶As Uhlig and Xu (1996) stress, the two main theoretical views with respect to cyclical movements in effort have opposite implications for employment fluctuations. Models of labour hoarding imply that effort co-moves positively with employment. Because labour input cannot be adjusted costlessly in the short run, firms react to shocks by varying the intensity of labour utilization. Thus, effort tends to increase in booms and decrease in recessions. In this context, effort movements allow firms to smooth the labour force over the cycle. By contrast, efficiency wage models that allow for adjustable effort predict a negative correlation between effort and employment, because increases in workers' performance are driven by increases in the threat of unemployment. In this context, effort movements amplify employment fluctuations.

⁷The ability to monitor workers affects the wage markup: as long as the detection probability is less than one, in equilibrium the wage rate will be above the Walrasian market clearing level and the surplus accruing to households from an employment relation will be positive.

⁸Among them, Lemieux, MacLeod and Parent (2009) document that consulting companies specializing in job evaluation, reward services and pay strategy, such as Hay Associates, Hewitt and Towers Perrin,

has been at the root of the extraordinary increase in reliance on pay-for-performance mechanisms since that time.⁹ My model suggests the following conjecture. In the 1970s firms' monitoring of workers' performance was poor. This reduced the risk of firing as a method of discipline, thus no performance-unemployment spiral was at work. In more recent years, firms' assessment of workers' performance has improved and possibly driven the economy in the region where the discipline device role of unemployment has been particularly strong. In this region, the high responsiveness of insiders' performance to unemployment helps to explain large and persistent employment movements. According to the model, further improvements in the ability to assess workers' performance, towards overcoming the asymmetric information and the ensuing moral hazard problem, can smooth the cyclical movements in labour input.

Investment in the development of better performance appraisal systems can be thought of as an important part of the paradigm shift towards efficiency gains, blamed by Yellen (2010) for the frustratingly slow rate of job creation. This does not rule out the potential importance of other mechanisms. I leave the empirical assessment of the importance of the channel highlighted by this model to future research.

This paper is organized as follows. Section 2 reviews the related literature. Section 3 presents the model. Section 4 provides the impulse response functions and describes the economic intuitions. It also provides a comparison of the effects when workers' effort is perfectly observable. Section 5 concludes.

2 Related literature

A number of recent works address the delayed recovery of employment following the last three recessions.

Shimer (2010) points to wage rigidities, and studies the transitional dynamics in an

have experienced extraordinary growth since the 1980s, and report, as examplars of this trend, the extraordinary surge in the sales of SAP, a major supplier of software used to monitor employee performance, from DM150 m in 1985 to \$8.8 bn today.

⁹As shown by Lemieux, MacLeod and Parent (2009), the enhancement in the quality of performance measures can explain the widespread evidence of firms' higher reliance on performance related compensation schemes as well as higher performance-pay sensitivity, because the availability of good measurement systems is a necessary condition for the effectiveness of a pay-for-performance system (Baker, 1992). The increased incidence of performance-pay has had a number of important implications at both the macro and micro levels: it accounts for shifts in the structure of correlations among US macro variables (Nucci and Riggi, 2011); leads to higher firms' productivity (Lazear, 2000); causes substantial increases in the volatility of wages relative to output (Champagne and Kurmann, 2011); and explains a large part of the rise in wage inequality in the US and why changes in US inequality have been concentrated increasingly at the top levels in the wage distribution (Lemieux, MacLeod and Parent, 2009).

economy that has lost some of its capital stock, when wages hold steady along a balanced growth path. The model describes a search economy in which firms adjust the labour input only along the extensive margin (number of workers). My paper shares a number of features with Shimer's (2010) study. First, as in Shimer (2010), I consider the effects of an adverse shock to the capital accumulation process. Second, the conclusion reached by Shimer (2010) is that, following the adverse shock, a model with flexible wages never generates a large decline in employment. The latter is indeed much less volatile than output. In order to get a large quantitative decline in employment, one has to assume rigid real wages, that do not fall after the shock. Consistent with this conclusion, the large movements in unemployment in the model I propose are associated with real wage rigidity, which, however, is generated endogenously by efficiency wages, rather than imposed as instead it is in Shimer (2010). To clarify this point, Shimer (2010) stresses that his model does not explain why wages are rigid, since the flexible wage path is also consistent with individual rationality. Instead, it makes the point that conditional on a path of individually rational wages, the model is consistent with jobless recoveries if and only if wages are rigid. By contrast, the moral hazard model I develop explains why rigid real wages may prevail in equilibrium.

Another explanation for the weak employment growth observed in the aftermath of the last three recessions relies on sectoral adjustments and has its roots in a large body of research, stemming from Lilien's (1982) influential contribution, aimed at exploring whether sectoral, rather than aggregate, shocks are responsible for fluctuations in the unemployment rate. Along this line of research, Andolfatto and MacDonald (2004) ascribe jobless recoveries to the uneven impact of technology innovations on different sectors along with slow sectoral adjustments in the labour market. The empirical evidence is not conclusive about whether the pace of sectoral reallocations has been considerably higher over the last three recoveries and whether sectoral adjustments played a crucial role in driving the stalled growth in jobs.¹⁰

The contraction of credit may have played crucial role in driving unemployment move-

¹⁰The historical decomposition provided by Loungani and Trehan (1997), using a VAR estimation and data up to 1995, reveals that sectoral shocks were most important during the 1974–75 recession, but explain only a modest part of the rise in unemployment in the 1990 recession. Groshen and Potter (2003) report data suggesting that most of the jobs added during the recovery following the 2001 recession were new positions in other industries rather than rehires, thus suggesting that the sluggishness observed in the job market may be attributed to the relocation of workers. Chen, Kannan, Loungani and Trehan (2011) document that sectoral shocks account for about half of the increase in the long duration unemployment rate that took place during the last recession (2007-2009), concluding that, in this, the recent global recession is similar to the recession of 1973-75, as sectoral shocks appear to have played a large role at that time as well.

ments, especially during the last global crisis, by inducing employers to cut hiring because of financial difficulties. However, Monacelli, Quadrini and Trigari (2011) argue that this standard credit channel cannot account for the delayed labour market recovery once the liquidity has rebounded and shed light on another channel through which de-leveraging can drive persistent increases in unemployment. Their model builds on empirical studies showing that firms may use financial leverage strategically in order to contrast the bargaining power of workers: wage bargaining induces preference for debt on the part of firms, because higher debt reduces the net bargaining surplus, which in turn reduces the wages paid to workers. When an adverse shock cuts the availability of credit for employers, de-leveraging raises workers' bargaining power and this leads firms to create fewer jobs. As long as the credit contraction is persistent the negative effect on unemployment is long-lasting.

Van Rens (2005) shows that jobless recoveries can be accounted for by substitution between organizational capital and labour. His model features two types of labour inputs: regular productive tasks and organizational capital accumulated by workers performing organizational tasks. When the recovery starts, hiring costs make it profitable to move workers from organizational tasks to productive activities, allowing firms temporarily to increase production without hiring extra workers.

By estimating a new-Keynesian model, Gali, Smets and Wouters (2011) come to the conclusion that the slower recoveries are not due to structural changes in the US economy, but rather due to the different contribution of the "recovery shocks" (i.e. shocks experienced during the recoveries themselves rather than shocks that originated during the preceding recessions), which were on average largely favorable in the pre-1990 period and have been somewhat adverse during the post-1990 period.

Gordon (2010) interprets the recent jobless recoveries as the result of two complementary phenomena: the "disposable worker" and the lagged benefits from the substantial boom of investment in information and communication technology (ICT) in the late 1990. First, the exceptionally slow recovery of the labour market appears to reflect many of the same causal factors that increased inequality and boosted the top decile income share over the last 25 years in the US (Piketty and Saez, 2006; Atkinson and Piketty, 2006): a declining minimum wage, weakening unions, an increase in imported goods, an increase in the number of immigrants for unskilled jobs and, above all, the shift in executive compensation toward stock options, which has exacerbated firms' emphasis on maximizing shareholder value. According to Gordon (2010), during the latest recession/recovery periods, the interplay between increased reliance of executive compensation on stock options and the collapse of profits and of the stock market created unusual pressure on corporate managers to cut costs and reduce employment, in a context in which such "savage corporate cost cutting" was no longer constrained by the countervailing power of labour.¹¹ Second, the aggressive cost cutting and the associated stellar productivity growth was made possible by the delayed effects of the ICT revolution that occurred in the latter part of the 1990s, a hypothesis that is upheld by Bernanke (2003).

All these factors have likely contributed to the slowness of the labour market revival after the last three downturns. My explanation is one of a complementary rather than a substitute mechanism.

My paper is related also to studies on shocks to the capital accumulation process. Many works show that these innovations are important drivers of aggregate fluctuations (Barro 2006, Fischer 2006, Justiniano and Primiceri 2008, Justiniano et. al 2010, Liu et al 2010, Furlanetto and Seneca 2011). These shocks, which take different forms in the literature,¹² have been used to model the global crisis in the late 2000s (Gertler and Karadi, 2011; Gertler and Kiyotaki, 2011) and the subsequent slow recovery of the labour market (Shimer 2010). The reason: shocks to the capital accumulation process can be seen as a reduced-form way of capturing financial frictions which impact on the supply of new capital. As argued by Shimer (2010), while in reality no capital was destroyed during the recession, some investments made prior to the global downturn turned out to be worth much less than originally forecast and the global crisis accelerated the contraction of some industries whose capital was ill-suited to the current economic environment and not easily adaptable to other purposes. Shimer (2010) stresses that this is analogous to a shock to the size of the capital stock. In addition, the 1990-91, 2001, and 2007-09 recessions were driven by different economic turmoils. An adverse shock to the capital accumulation process reflects a disturbance that renders a part of the capital stock economically obsolete and thus appears well suited to stylizing the three recent crises, without taking a stand on the specific nature of the downturn.

Finally, my paper relates to the prominent literature on efficiency wages, which treats unemployment as a real phenomenon resulting from incentive problems in the labour

¹¹The evidence provided by Oliner, Sichel and Stiroh (2007) confirms that productivity growth after 2000 has been boosted by industry restructuring and cost cutting in response to profit pressures. They document that those firms that had experienced the largest declines in profits between 1997 and 2002 also exhibited the largest declines in employment and the largest increases in productivity.

¹²Given the capital accumulation equation $K_{t+1} = \xi_{q,t} \left[(1 - \varrho_t) K_t + \xi_{I,t} \varphi \left(\frac{I_t}{K_t} \right) K_t \right]$, the shock to the capital accumulation process may take the form of a depreciation shock ϱ_t as in Shimer (2010), an investment specific technology shock $\xi_{I,t}$ as in Justiniano and Primiceri (2008) and Justiniano et al. (2010), or a shock to the quality of capital $\xi_{q,t}$ as in Gertler and Kiyotaki (2011)

market. Much of this literature employs a static partial equilibrium approach in order to investigate the determinants of the steady-state level of unemployment. Relevant contributions towards understanding labour market dynamics outside of the steady state are those of Kimball (1994), Woodford (1994), Danthine and Donaldson (1995), Gomme (1999) and Alexopoulos (2004). My work takes another step in this direction.

3 The model

I consider an economy which grows along a balanced growth path. The model integrates search and matching frictions in the labour market - captured through hiring costs increasing with labour market tightness - and efficiency wages. Because workers' performance is imperfectly observable, firms pay above-market wages in order to induce workers to exert the desired amount of effort. The latter is adjustable on a continuous scale. In this context, I look at the effects of an adverse and transitory shock to the capital accumulation process.

3.1 Preferences

The relevant decision unit is the infinitely-lived representative household with a continuum of members represented by the unit interval. State-contingent securities offer workers full insurance against differences in their specific income.¹³ The household's objective function is consistent with a balanced growth path¹⁴ and is given by:

$$\mathbb{E}_0 \sum_{t=0}^{\infty} \beta^t \left[\log C_t - \int_0^{N_t} \widehat{\mathcal{E}}_{i,t} \mathrm{d}i \right]$$
(1)

¹³The household assigns equal consumption to all members in order to share consumption risk within the family. Indeed, in the presence of unemployment risk, differences in consumption levels between employed and unemployed workers might emerge. The full income insurance scheme avoids this issue and implies that the income of an unemployed person is the same as the income of an employed member who is not detected shirking. As highlighted by Alexopoulos (2004), this insurance scheme, which is the one that the family would choose to maximize its expected utility, requires the following additional assumptions in order to guarantee that individuals will always accept employment: the family can observe which members receive job offers and punishes individuals who turn down a job offer by making them ineligible for any intra-family transfer.

¹⁴I abandon Shapiro and Stiglitz's (1984) assumption of a constant marginal utility of income, in order to have wealth effects on equilibrium wage determination. This allows a balanced growth path along which unemployment is trendless.

where $\beta \in (0, 1)$ is the discount factor, C_t denotes consumption, $N_t \in [0, 1]$ is the fraction of household members who are employed, and $\widehat{\mathcal{E}}_{i,t} \geq 0$ is the amount of effort exerted by each employed household member *i*.

Workers' performance is imperfectly observable and shirkers are detected with probability d. The term "shirker" stands for a worker that provides a level of effort $\widehat{\mathcal{E}}_{i,t}$ lower than the amount requested by firms \mathcal{E}_t . Detected shirkers forgo payment of the wage and are dismissed.¹⁵ Denoting with N_t^{ε} the fraction of household members who are employed and do not shirk on the job and with N_t^s the fraction of shirkers, we have $N_t = N_t^s + N_t^{\varepsilon}$.

The budget constraint is given by:

$$C_t + I_t = \int_{0}^{N_t^{\varepsilon} + (1-d)N_t^{s}} W_{i,t} \mathrm{d}i + R_t^k K_t + \Pi_t$$
(2)

where I_t denotes investment expenditures, $W_{i,t}$ is the real wage accruing to household member *i* and Π_t are firms' profits. I denote with K_t the household's capital holdings, rent to firms at the (real) rental cost R_t^k .

The capital accumulation equation is:

$$K_{t+1} = (1 - \varrho_t) K_t + \varphi \left(\frac{I_t}{K_t}\right) K_t$$
(3)

where ϱ_t is a stochastic variable which denotes the time-varying depreciation rate of capital. I assume

$$\varrho_t = \overline{\varrho} e^{v_t} \tag{4}$$

where v_t is an i.i.d. shock with mean 0 and standard deviation 1 and $\overline{\varrho}$ is the mean depreciation rate. A positive shock to ϱ_t renders some of the capital stock unproductive. The term $\varphi\left(\frac{I_t}{K_t}\right)K_t$ captures capital adjustment costs, which determines the change in the capital stock induced by investment spending. As in Gali et al. (2007), I assume:

$$\varphi' > 0$$
, and $\varphi'' \le 0$, with $\varphi'(\frac{\widetilde{I}}{\widetilde{K}}) = 1$ and $\varphi(\frac{\widetilde{I}}{\widetilde{K}}) = \frac{\widetilde{I}}{\widetilde{K}}$ (5)

where $\frac{\tilde{I}}{\tilde{K}}$ is the steady state investment to capital ratio. From the first order conditions

¹⁵The assumption that the cost imposed on disciplined workers includes payment of no wage, alongside dismissal, is based on Woodford (1994) and on Kimball (1994). As pointed out by Woodford (1994), Kimball's (1994) continuous-time model does not represent the infinitesimal-period-length of a model in which the only penalty for detected shirkers is termination but rather it is the limit of a model in which the current period wage is also withheld.

one obtains that:

$$Q_t^{Tobin} = \mathbb{E}_t \left\{ \beta \frac{C_t}{C_{t+1}} \left[R_{t+1}^k + Q_{t+1}^{Tobin} \left((1 - \varrho_t) + \varphi_{t+1} - \left(\frac{I_{t+1}}{K_{t+1}} \right) \varphi_{t+1}' \right) \right] \right\}$$
(6)

$$Q_t^{Tobin} = \frac{1}{\varphi'\left(\frac{I_t}{K_t}\right)} \tag{7}$$

where Q_t^{Tobin} is the (real) shadow value of capital in place, i.e., Tobin's Q, and $\varphi_{t+1} = \varphi\left(\frac{I_{t+1}}{K_{t+1}}\right)$ and $\varphi'_{t+1} = \varphi'\left(\frac{I_{t+1}}{K_{t+1}}\right)$.

3.2 The labour market

In equilibrium profit maximizing firms will offer a wage that ensures workers will not shirk on the job. Employment evolves according to:

$$N_t = (1 - \delta) N_{t-1} + X_t U_t$$
(8)

where $\delta \in (0, 1)$ is an exogenous separation rate,¹⁶ U_t denotes the size of the pool of jobless individuals available for hire at the beginning of period t, and X_t is the job finding rate, defined by

$$X_t \equiv \frac{H_t}{U_t},\tag{9}$$

where H_t denotes aggregate hiring. Workers are immediately productive in the period when they are hired¹⁷. Firms incur a cost to hire new workers. As in Blanchard and Gali (2010), the cost per hire (G_t) is taken as given by each firm and is increasing with labour

¹⁶The assumption of an acyclical separation rate is based on recent works proposing a dominant role of the job finding rate in explaining workers' flows. The standard view that recessions are periods characterized primarily by high job loss rates (see Blanchard and Diamond, 1990) has been contradicted by the last three downturns, which have not seen a wave of job losses: during the last two decades the sharp surges in unemployment have been due not to spikes of layoffs but rather to the fact that, once unemployed, workers' chances of finding jobs have fallen sharply. Hall (2005) concludes that in the modern US economy unemployment rises because it is hard to find a job and not because an unusually large number of people are thrown into unemployment. This view has been recently corroborated during the last recession (Elsby, Hobijn, and Sahin, forthcoming). Shimer (2005, 2007) observes that over the last two decades the separation probability is acyclical, whereas the job finding probability is strongly procyclical. See Barnichon (2011) for cautions about the assumption of an acyclical separation rate.

¹⁷This timing convention is analogous to many papers. Among them, Blanchard and Gali (2010), Gertler, Sala and Trigari (2008), Faia (2009).

market tightness X_t :¹⁸

$$G_t = \gamma^t B X_t^\eta \tag{10}$$

where γ is the deterministic growth rate,¹⁹ B is a positive constant and $\eta \geq 0$ measures the elasticity of hiring costs to labour market conditions.

3.3 Firms

I assume a continuum of identical firms $j \in [0, 1]$, which operate in a competitive market and produce a homogeneous consumption good with the following technology:

$$Y_{jt} = \left(\gamma^t \int_0^{N_{jt}} \widehat{\mathcal{E}}_{jit}^{\varkappa} \mathrm{d}i\right)^{\alpha} K_{jt}^{1-\alpha}$$
(11)

where γ is the growth rate, capturing trend productivity growth, $\alpha \in (0, 1)$ and $\varkappa \in [0, 1]$.

Firms do not perfectly observe workers' performance. As a consequence, in order to induce workers to exert the desired amount of effort, they must offer them a wage that satisfies the incentive compatibility constraint.

The incentive compatibility constraint

The household chooses the level of effort provided by each employed member and, because a worker is considered a shirker for any level of effort $\widehat{\mathcal{E}}_{i,t} < \mathcal{E}_t$, the household maximizes its utility by choosing $\widehat{\mathcal{E}}_{i,t} = \mathcal{E}_t \,\forall i$, if the family wants its members to exert any effort, and $\widehat{\mathcal{E}}_{i,t} = 0$, otherwise (see Alexopoulos, 2004). The former case holds if the incentive compatibility constraint is satisfied.

The incentive compatibility constraint can be expressed as follows: the marginal value expressed in terms of the consumption goods accruing to the household generated by an employed non-shirker member $(\mathcal{V}_t^{N,\varepsilon})$ should be at least as great as that generated by an employed shirker member $(\mathcal{V}_t^{N,\varepsilon})$: $\mathcal{V}_t^{N,\varepsilon} \geq \mathcal{V}_t^{N,S}$. In equilibrium, the constraint holds with equality and defines the *no shirking condition*, which is the equilibrium relationship between performance supply and wages. One has:

¹⁸As in Blanchard and Gali (2010), X_t is both an index of labour market tightness and, from the viewpoint of the unemployed, the probability of being hired in period t.

¹⁹As in Blanchard and Gali (2010), the hiring cost is assumed to grow with productivity in order to rule out that productivity improvements can affect the cost of hiring relative to the cost of producing.

$$\mathcal{V}_{t}^{N,\varepsilon} = W_{t} - C_{t}\mathcal{E}_{t} + \\ + \mathbb{E}_{t}\beta \frac{C_{t}}{C_{t+1}} \left\{ \left[1 - \delta \left(1 - X_{t+1} \right) \right] \mathcal{V}_{t+1}^{N} + \delta \left(1 - X_{t+1} \right) \mathcal{V}_{t+1}^{U} \right\}$$
(12)

$$\mathcal{V}_{t}^{N,S} = (1-d)W_{t} + \\ + \mathbb{E}_{t}\beta \frac{C_{t}}{C_{t+1}} \left\{ \left[1 - (\delta+d)\left(1 - X_{t+1}\right)\right] \mathcal{V}_{t+1}^{N} + (d+\delta)\left(1 - X_{t+1}\right) \mathcal{V}_{t+1}^{U} \right\}$$
(13)

where V_t^U denotes the marginal value expressed in terms of the consumption goods accruing to the household, generated by an unemployed member. As in Shapiro and Stiglitz (1984) job histories do not matter: unemployed workers who were fired for shirking are indistinguishable from other unemployed workers, thus they all face the same probability of being hired in a given period. The *no shirking condition* can be written as:

$$\mathcal{E}_{t} = dU_{c,t} \left[W_{t} + \mathbb{E}_{t} \beta \frac{C_{t}}{C_{t+1}} \left(1 - X_{t+1} \right) \mathcal{S}_{t+1}^{H} \right]$$
(14)

where $U_{c,t} = \frac{1}{C_t}$ denotes the marginal utility of consumption and $S_t^H \equiv \mathcal{V}_t^N - \mathcal{V}_t^U$ defines the surplus accruing to the household from an established employment relationship. Equation (14) states that in equilibrium the level of effort that workers exert depends on the loss they would incur were they detected as shirkers, weighted by the probability of being detected (d). This loss is given by the sum of two components: the real wage they would forgo, were they detected, plus the expected present discounted value of the future surplus from an employment relation that they would forgo were they detected and not re-employed, which happens with probability $d(1 - X_{t+1})$.

Unemployment is a method of discipline because, as in Shapiro and Stiglitz (1984), if a worker is fired, he will not immediately find another job. This is captured by the term $(1 - X_{t+1}) = (1 - \frac{H_{t+1}}{U_{t+1}})$, which measures the expected probability of not finding another job: the higher the prevailing level of unemployment, the longer the expected spell of unemployment and the higher the level of effort that workers are willing to exert. Thus the threat of firing, implying a spell of unemployment, deters shirking. The strength of the discipline device role of unemployment can be quantified as follows:

$$\mathcal{I}_t \equiv \frac{\partial \mathcal{E}_t}{\partial (1 - X_{t+1})} = U_{c,t} d\mathbb{E}_t \beta \frac{C_t}{C_{t+1}} \mathcal{S}_{t+1}^H$$
(15)

In order to eliminate \mathcal{S}_{t+1}^H in equation (14), \mathcal{V}_t^U can be written as:

$$V_t^U = \beta \mathbb{E}_t \left\{ \frac{C_t}{C_{t+1}} \left[(1 - X_{t+1}) \,\mathcal{V}_{t+1}^U + X_{t+1} \mathcal{V}_{t+1}^N \right] \right\}$$
(16)

Subtracting (16) from (12) and using (14) yields:

$$\mathcal{S}_t^H = \delta W_t + \left(\frac{1 - \delta - d}{d}\right) C_t \mathcal{E}_t \tag{17}$$

The no shirking condition can be rewritten as follows:

$$C_t \mathcal{E}_t = dW_t +$$

$$+ \mathbb{E}_t \beta_{\frac{C_t}{C_{t+1}}} (1 - X_{t+1}) \begin{bmatrix} d\delta W_{t+1} + \\ + (1 - \delta - d) C_{t+1} \mathcal{E}_{t+1} \end{bmatrix}$$
(18)

or, denoting $MRS_{(C,N)_t} \equiv C_t \mathcal{E}_t$, in the following way:

$$W_{t} = \frac{MRS_{(C,N)t}}{d} - \mathbb{E}_{t}\beta \frac{C_{t}}{C_{t+1}} \left(1 - X_{t+1}\right) \left[\delta W_{t+1} + \frac{(1 - \delta - d)}{d} MRS_{(C,N)t+1}\right]$$
(19)

Note that in the stationary steady state²⁰ the no shirking condition implies that

$$\widetilde{W} = \widetilde{C} \mathcal{E} \underbrace{\frac{\left[1 - (1 - \delta - d)\beta(1 - x)\right]}{d\left[1 + \delta\beta(1 - x)\right]}}_{\text{steady state wage markup}}$$
(20)

where \sim indicates that the variable has been detrended by the level of technology γ^t and x denotes the steady state job finding rate. Consistently with the efficiency wage literature (and in particular with Shapiro and Stiglitz, 1984), the steady state critical wage must be higher when:

- the desired level of effort is higher
- the probability of being caught shirking (d) is lower
- the exogenous separation rate (δ) is higher (that the workers are going to lose their job soon anyway increases the incentive to shirk)
- the job finding rate (x) is higher (that the workers will be re-employed soon after losing their job increases the incentive to shirk)
- the discount factor (β) is lower (if future benefits that workers would forgo if they were detected and not re-employed- matter less, the incentive to shirk is higher).

²⁰The economy grows along a balanced growth path: trend productivity growth determines trend increases in output, consumption, wages, capital and investment, without affecting employment and workers' effort. The stationary representation of the model can be obtained by detrending the former group of variables.

The firm's problem

Because it is not profitable for firms to hire shirking workers, in equilibrium profit maximizing firms will offer a wage that ensures workers will not shirk on the job, i.e. $\hat{\mathcal{E}}_{i,t} = \mathcal{E}_t$ $\forall i$ and $N_t = N_t^{\mathcal{E}}$. This leads to the following expression of the firm's period t problem:

$$\max_{\{W_{j,t} \; H_{j,t} \; N_{j,t} \; \mathcal{E}_{j,t} \; K_{j,t}\}} \mathbb{E}_{t} \sum_{k=0}^{\infty} Q_{t,t+k} \left[\begin{array}{c} \left(\gamma^{t+k} N_{jt+k} \mathcal{E}_{jt+k}^{\varkappa}\right)^{\alpha} K_{j,t+k}^{1-\alpha} + \\ -W_{j,t+k} N_{j,t+k} - \gamma^{t+k} B X_{t+k}^{\eta} H_{j,t+k} - R_{t+k}^{k} K_{j,t+k} \end{array} \right]$$
(21)

subject to (8) and the incentive compatibility constraint (18), where $Q_{t,t+k} = \beta^k \frac{C_t}{C_{t+k}}$ is the stochastic discount factor. Assuming that firms take private sector expectations as given to solve the optimization problem would require wages and effort at time (t+1) to be a function only of exogenous state variables. Because this is not the case, since employment is an endogenous state variable, the equilibrium I consider is a timeless-perspective commitment in which firms internalize the effects of their choices on workers' expectations. Denoting with $MPE_t \equiv \frac{\alpha \times Y_t}{\mathcal{E}_t}$ the marginal product of effort and with $MRS_{(C,\mathcal{E})t} \equiv C_t N_t$ its marginal disutility in terms of consumption, the first order conditions for the firm's problem imply:

$$W_{t} = \alpha Y_{t} / N_{t} - B \gamma^{t} X_{t}^{\eta} + (1 - \delta) \mathbb{E}_{t} Q_{t, t+1} B \gamma^{t+1} X_{t+1}^{\eta}$$
(22)

$$MRS_{(C,\mathcal{E})t} = dMPE_t + \frac{C_t}{C_{t-1}} \left(1 - X_t\right) \left[MRS_{(C,\mathcal{E})t-1} \left(1 - \delta - d\right) + d\delta MPE_{t-1}\right]$$
(23)

$$R_t^k = (1 - \alpha) \frac{Y_t}{K_t} \tag{24}$$

Equation (22) is the *pseudo*-labour demand, that gives the optimal hiring policy. According to (22) each period the firm hires workers up to the point where the marginal product of labour $(\alpha Y_t/N_t)$ equals the cost of a marginal worker. The latter is given by the real wage, the hiring cost and the discounted savings in future hiring costs resulting from having to hire $(1 - \delta)$ fewer workers in the following period.

Equation (23) is a sluggish relation between the marginal product of effort and its marginal disutility expressed in terms of consumption, which can be read as giving the performance-employment frontier. Note that (23) is a backward looking equation: because

the incentive compatibility constraint faced by firms is a forward looking constraint, in a commitment equilibrium the implied policy does not change much over time. As long as worker's performance is imperfectly observable (d < 1), the level of effort will not be determined efficiently, and, in steady state, its marginal product will be greater than its marginal disutility expressed in terms of consumption:²¹ $\frac{\widetilde{MPE}}{\widetilde{MRS}_{(C,\mathcal{E})}} = \frac{[1-(1-\delta-d)(1-x)]}{[1+\delta(1-x)]d} > 1$. Besides $\frac{[1-(1-\delta-d)(1-x)]}{[1+\delta(1-x)]d}$ is monotonically strictly decreasing with d, meaning that increases in the ability to assess workers' performance are associated with increases in the steady state level of effort (see Appendix C).

Equation (24) gives the demand for capital.

3.4 Aggregate resource constraint

Market clearing requires:

$$Y_t = C_t + G_t H_t + I_t \tag{25}$$

Appendix A provides the steady state relations and the complete log-linear model in stationary form.

4 Capital destruction and unemployment

4.1 Unobservable effort

Suppose that the economy is at a steady state and an unanticipated, transitory, one percentage point increase in the depreciation rate of capital destroys some of the capital stock. In this section, I show that the effects on unemployment depend crucially on the extent to which workers' performance is observable.

The rationale for studying the effects of the shock conditional on d is twofold. On the one hand, I will show that the extent to which workers' performance is observable is key for determining the strength of the discipline device role of unemployment. In addition, as clarified in section 4.2, when d tends to 1 the model's results become comparable to those obtained by Shimer (2010), although my model features two margins for adjusting firms' effective labour input (employment and effort).

The time period is a quarter. I use conventional values for the discount factor ($\beta = 0.998$), the mean depreciation rate ($\overline{\rho} = 0.025$) and the share parameter on capital ($1-\alpha =$

²¹When d=1, then $\widetilde{MPE} = \widetilde{MRS}_{(c,\varepsilon)}$.

1/3).²² The elasticity of the investment-capital ratio with respect to Tobin's Q, μ , is assumed to be equal to 1 (as in King and Watson 1996 and Gali et al. 2007). I set $\varkappa = 0.9$, in order to have additional diminishing returns to effort. I fix $\delta = 0.105$, based on the observation that jobs last about two and a half years (Shimer 2005 and Gertler et al. 2008). Based on Shimer (2005) finding that the average exit probability from unemployment to employment in the US is 0.34 per month, I set x = 0.7 (as in Blanchard and Gali 2010).²³ I follow Blanchard and Gali (2010) and set $\eta = 1.^{24}$ I fix $\gamma = 1.004$, consistent with a steady state value for productivity growth of 0.4% on a quarterly basis. Finally, I assume that the ratio of the cost per hire to the quarterly real wage $(h_w \equiv \frac{Bx^{\eta}}{\widetilde{W}})$ is 0.15, meaning that the cost per hire is 4% of the annual wage. As shown in Appendix A, hiring costs represent a fraction $h_y \equiv \frac{Bx^{\eta}H}{\widetilde{Y}} = h_w \frac{\alpha\delta}{1+h_w[1-(1-\delta)\beta]}$ of GDP. According to my calibration, that fraction equals one per cent of GDP, which is a plausible upper bound, in line with what is assumed in Blanchard and Gali (2010). Appendix B shows that, under the calibration discussed above, wages have a negligible probability of falling outside the bargaining set, i.e. the range of wage levels above the workers' reservation wage and below the firms' reservation wage.

Impulse response functions

Figure 2 reports the impulse response functions (IRFs) of employment, effort, output, investment, real wage, capital, consumption and capital rental cost for different values of d. All the IRFs, except employment and effort, are expressed relative to trend. The following emerges:

• An adverse shock to the size of the capital stock leads to recession, regardless of the firms' ability to evaluate workers' performance. Capital destruction causes proportional declines in output, investment and consumption relative to trend. These three variables display roughly the same volatility.

²²Note that, because of labour market frictions, α does not measure labour share, which is given in the model by: $\frac{\alpha}{1+h_w[1-(1-\delta)\beta]}$, where h_w denotes the ratio of the cost per hire to the quarterly real wage. Because this share is just below α , I simply follow convention by setting $\alpha = 1 - (1/3)$. Under this calibration the labour share is 0.66, which is in line with US macro data.

²³As in Blanchard and Gali (2010), the equivalent quarterly rate is computed as $x = x_m + (1 - x_m)x_m + (1 - x_m)^2 x_m$, where x_m is the monthly job finding rate.

²⁴As stressed by Blanchard and Gali (2010), assuming a matching function of the form $H = ZU^{\varsigma}V^{1-\varsigma}$, we have $\frac{V}{H} = Z^{\frac{1}{\varsigma-1}} \left(\frac{H}{U}\right)^{\frac{\varsigma}{1-\varsigma}}$. Thus, the parameter η corresponds to $\frac{\varsigma}{1-\varsigma}$ in the standard Diamond, Mortensen, Pissarides model and $\eta = 1$ is consistent with estimates of ς , which typically are close to 0.5.

- There is a negative correlation between employment and workers' performance, i.e. effort and employment move in opposite directions as the shock hits the economy: when employment falls, effort rises and vice-versa. In addition, the model delivers two sets of outcomes, depending on the firms' ability to evaluate workers' performance. As long as d is less than a certain threshold value (d_*) , employment rises and effort declines in the face of the capital depreciation shock; and vice-versa, when d is larger than a certain threshold value (d_*) the shock leads to a drop in employment and an increase in effort. When $d = d_*$ the model falls in the region of indeterminacy. Given the calibration discussed above, $d_* \approx 0.8$. To give a sense of magnitude, d = 0.8 implies a steady state wage markup over the reservation wage of 18%.
- The firms' ability to assess workers' performance affects not only the sign but also the size of the employment and effort movements. As long as $d < d_*$, the larger dis, the larger the variations in employment and effort. By contrast when $d > d_*$, the larger d is, the smaller the variations in employment and effort.
- Real wages fall and the negative responses of the real wage associated with positive responses of employment are larger (in absolute value) than those associated with negative responses of employment. Figure 3 focuses on the relationship between real wage and labour input movements, showing that the smaller the drop in the real wage the more positive the response of workers' performance (Panel A) and the more negative the response of employment (Panel B). Accordingly, when $d \rightarrow d_*^+$ the dramatic drop in employment and the sharp rise in effort come together with rigid real wages, which roughly do not move.

To sum up, Figure 4 collects the IRFs of output, investment, consumption and real wage relative to trend, and of employment and effort when d = 0.85, a value which is slightly larger than d_* (and which implies a steady state wage markup over the reservation wage of 12.52%). Employment and output, consumption and investment relative to trend fall with the impact of the shock and have nearly the same volatility. Workers' performance booms and the real wage does not move. Thereafter consumption, investment and output grow at around their usual balanced growth rate γ , with employment persistently at the depressed level and effort persistently at the high level: the adverse shock rendering some of the capital stock unproductive generates a jobless recovery driven by a boom in workers' performance. Figure 5 shows the dynamic simulation of employment and undetrended GDP and labour productivity in levels. Importantly, the same scenario for employment,

output, investment and consumption is obtained in Shimer's (2010) search model, in which firms adjust labour input only along the extensive margin, by imposing that the real wage stays fixed. Similarly here, despite sizeable movements in unemployment, the real wage remains roughly constant but as a result of the forces at stake in the economy.

Let me stress that, as in Shimer (2010), I use the term "jobless recovery" in order to refer to a business cycle phase during which GDP growth turns positive while employment stays depressed. However, note that, as in Shimer (2010), following the adverse capital depreciation shock, the level of GDP remains persistently below its potential.

Economic intuitions

I cannot solve for the equilibrium analytically, but the driving forces behind the results are clear-cut.

Unemployment acts as a threat that motivates workers to devote effort to their works. This is the reason behind the negative correlation between effort and employment, shown in Figure 2. Let's consider again the incentive compatibility constraint:

$$W_t = \frac{\mathcal{E}_t}{dU_{c,t}} - \mathbb{E}_t \beta \frac{C_t}{C_{t+1}} \mathcal{S}_{t+1}^H \left(1 - X_{t+1}\right)$$
(26)

which represents the supply-wage as a positive function of the job finding rate.²⁵ The positive slope reflects the discipline device role of unemployment: for any given level of effort, the higher the job finding rate, the higher will be the incentive compatible wage. The intersection of the *pseudo labour demand* (22) and the no-shirking condition (26) is a downward sloping locus, mapping the quantity of labour into the effort level such that the demand wage equates with the incentive compatible wage.²⁶

Two effects ensue from the capital depreciation shock. First, workers are poorer. Because preferences are such that the marginal utility of consumption $(U_{c,t})$ increases as

²⁵As shown in section 3.3 $(1 - X_{t+1}) = \left(1 - \frac{H_{t+1}}{U_{t+1}}\right)$ is the probability of not finding a job and remaining unemployed.

²⁶In the Shapiro and Stiglitz (1984) model workers can provide either a zero level of effort, or some fixed positive level $\overline{\mathcal{E}} > 0$. Hence, the no-shirking condition is a curve, drawn for $\overline{\mathcal{E}}$, and the intersection between the aggregate demand for labour and the no-shirking condition gives the equilibrium. Here, because effort is allowed to vary over the cycle, the no-shirking condition is a bunch of curves, each of which drawn for a different value of effort. The intersection between this bunch of curves and labour demand is a downward sloping locus, mapping the quantity of labour into the effort level such that the demand wage equates with the incentive compatible wage. The equilibrium levels of employment and effort occur where this locus intersects the desired relationship between effort and employment given by (23), that is the performance-employment frontier relating the marginal product of effort and its marginal disutility expressed in terms of consumption.

consumption falls, the reservation wage decreases and with it the wage required to deter shirking. Hence, this wealth effect implies that, for any given level of effort, the level of employment satisfying the incentive compatibility constraint increases. Second, labour productivity falls, making the pseudo labour demand shift to the left. Hence, this effect implies that, for any given level of effort, the level of employment satisfying the incentive compatibility constraint decreases. Both movements lead to a drop in the equilibrium wage level, while the sign of the response of employment is ambiguous.

Why is d crucial in determining the sign of the labour input response? The incentive compatibility constraint (26) shows that the former effect will be stronger with lower value of d. Intuitively d affects the sensitivity of the incentive compatible wage to the reservation wage: the smaller the detection probability, the higher the desired wage markup, thus the larger the change in the incentive compatible wage driven by a change in the reservation wage. As for the second effect, low values of d make the incentive compatibility constraint steeper, because workers' surplus S_{t+1}^H decreases with d. When the shift in labour demand traces out a steeper incentive compatibility constraint, the negative effect on employment is smaller than in the case of a fairly flat wage-supply schedule. Both considerations suggest that, as shown in Figure 2, given the other parameters, low values of d are more likely to be associated with positive responses of employment, whereas high values of d are more likely to be associated with negative responses of employment.

Figure 2 shows an important effect of d on the magnitude of employment and effort responses. In particular, as stressed above, when d is such that employment falls in the face of the capital depreciation shock (i.e. d is relatively high), the larger d is, the smaller the magnitude of employment and effort responses. Accordingly, large declines in employment and jobless recoveries emerge for values of d slightly larger than d_* . By contrast, when d is such that employment rises (i.e. d is relatively low), the larger d is, the larger the magnitude of employment and effort responses. The intuition hinges again on the threat of unemployment as a method of discipline.

Suppose that the economy is on its balanced growth path, there is a one-time capital depreciation shock and firms' ability to monitor workers is relatively high, so that they cut back on their employment. The increased threat of unemployment makes insiders more efficient, inducing them to work harder. The increased efficiency, given the effort-employment frontier, reduces firms' needs for workers, resulting in even greater unemployment. This further strengthens the threat and further boosts the effort, leading ultimately to amplified unemployment and performance dynamics. The opposite happens when d is relatively low. In this case the increased employment reduces the threat of firing, and makes insiders less efficient, inducing them to work less hard. Given the effortemployment frontier, the depressed efficiency increases firms' need for workers, resulting in lower unemployment, leading ultimately to amplified employment and performance dynamics. It follows that, both when employment rises and when it falls, the stronger the discipline device role of unemployment - i.e. the responsiveness of the level of effort that workers are willing to exert to changes in the expected probability of not finding a job - the larger the cyclical employment and effort movements. The effect that increases in d play on the effort responsiveness to unemployment is thus crucial to this economic intuition.

In section 3.3 I quantified the discipline device role of unemployment as follows:

$$\mathcal{I}_{t} \equiv \frac{\partial \mathcal{E}_{t}}{\partial (1 - X_{t+1})} = U_{c,t} d\mathbb{E}_{t} \beta \frac{C_{t}}{C_{t+1}} \mathcal{S}_{t+1}^{H}$$
(27)

Looking at (27), it can be seen that unemployment plays a weak role as a method of discipline not only when the detection probability d is close to zero - meaning that the threat of firing is weak - but also when it is close to one, because in this case the expected surplus accruing to the household from an existing employment relation S^H (that workers would lose if they were detected shirking and not re-employed) is close to zero. Indeed, if the detection probability tended to one the model would become a search and matching economy with zero workers bargaining power, the wage rate would be equal to the workers' reservation wage - since firms would have no reason to pay above the market wage - involuntary unemployment would be determined only by the presence of search and matching frictions. In other words, unemployment would play no role as a method of discipline. The discipline device role of unemployment is strong when the firms' ability to assess workers' performance is such that the probability of being detected shirking and joining the pool of unemployed is considerable and, at the same time, the loss workers would incur were they detected shirking is still considerable.

It follows that the increased ability of firms to monitor labour quality strengthens the role of unemployment as a discipline device as long as the decreased surplus does not counterbalance the increased fear of losing one's job. I cannot proceed analytically outside the steady state but, given the other parameters, this is clearly more likely when d is relatively small. Appendix C and Figure 6 clarify this point by illustrating the dependence of the discipline device role of unemployment, as quantified in (27) and evaluated in steady state, on d.

Summing up, when d is relatively high, so that firms cut back on their employment,

the strength of the discipline device role of unemployment decreases with d. This explains why the model delivers sharp declines in employment, proportional to declines in GDP relative to trend, when d is large enough to induce firms to cut back on employment and raise performance standard, but not large enough to dim the discipline device role of unemployment. In this case the large movements in unemployment are driven by the increased efficiency of insiders. Conversely, when d is relatively low, so that firms increase their employment, the strength of the discipline device role of unemployment increases with d. This explains why when $d < d_*$ the larger d is, the larger the cyclical movements in employment and effort.

Figures 2, 3B and 4 illustrate that drastically falling employment is synonymous with real wage rigidities. This important result is a direct consequence of the efficiency wage mechanism and can be explained as follows. As documented above, the drop in employment is associated with increases in workers' performance and the larger the drop in employment the larger the increase in effort. When effort increases, the marginal disutility of labour rises. This counterbalances the negative impact that the rise in the marginal utility of consumption has on the reservation wage (as the latter depends on $MRS_{(C,N)_t} \equiv \frac{1}{U_{c,t}} \mathcal{E}_t$). In consequence, the fall in workers' reservation wage, driven by the wealth effect, will be smaller, the larger the increase in effort (see Figure 3A). When the drop in employment and the specular rise in effort are proportional to the movements in consumption, the drop in the real wage is almost zero.

4.2 Observable effort

Shimer (2010) looks at the effects of an adverse shock to the capital accumulation process in a search model where firms can adjust labour input only along the extensive margin. The main conclusion is that a model with flexible wages never generates a large decline in employment. The latter is indeed much less volatile than output. In order to obtain a quantitatively large decline in employment, one has to assume rigid real wages, that do not fall after the shock. In this case the adverse shock leads to a proportional decline in employment and in output, consumption and investment relative to trend. Thereafter jobless recovery emerges because output, consumption and investment grow at around their usual growth rate γ while employment stays persistently at the depressed level.

In order to allow a comparison with Shimer's (2010) results, Figure 7 shows the impulse responses to a capital depreciation shock of employment and effort and of output, consumption, investment and real wages relative to trend implied by the model developed in section 3, when workers' performance is perfectly observable. Shimer's (2010) results are replicated in the model I put forward in this paper, which features the possibility to adjust labour input along both the extensive (employment) and the intensive (effort) margin.

The model is still calibrated to US data, as indicated in section 4.1. When wages are flexible employment declines by much less than output, consumption, wages and investment relative to trend. The positive movements in effort are muted as well as those in employment. Hence, the clearing of asymmetric information and moral hazard can smooth cyclical movements in effort and unemployment. When wages are assumed to be rigid,²⁷ the model produces an equal decline in employment and in output, consumption and investment relative to trend on the impact of the adverse shock. Thereafter, output, consumption and investment growth turns positive (at around γ) with employment persistently at the depressed level. It should also be noted that the behaviour of the economy in this case (Figure 7B) resembles the jobless recovery obtained under the assumption of efficiency wages when the discipline device role of unemployment is particularly strong (Figure 4). In the latter case however the response of effort is somewhat larger than in the case of observable effort.

5 Conclusions

Historically, as the economy emerged from a downturn, revival of GDP growth was accompanied by a prompt increase in employment. The 1990-91 recession broke this pattern and opened an era of jobless recovery, where the revival of GDP growth is associated with prolonged anemic growth in employment. Using an efficiency wage model with search and matching frictions, this paper has highlighted a channel that may have played a role in establishing this tendency.

The mechanism at stake hinges on incentive problems in the labour market. When an adverse shock to the capital accumulation process hits the economy and firms cut back on their personnel, the increased threat of being fired and of enduring long spells of unemployment can amplify the cyclical movements in workers' performance leading to large declines in employment and jobless recovery. Also, the model explains why rigid real wages may prevail in equilibrium. I have also shown that the clearing of asymmetric information and moral hazard can smooth cyclical movements in effort and unemployment. A

²⁷I assume that detrended real wage is constant at $\widetilde{W}_t = \widetilde{MRS}_{C,N}$ so that $\widetilde{w}_t = 0$.

general conclusion of this paper is therefore that incentive problems in the labour market help account for business cycle comovement.





NOTES. Civilian-Employment Population Ratio (%). Red shaded areas indicate NBERdated US recessions. Light blue shaded areas indicate jobless recoveries. Source: Bureau of labour Statistics.



Figure 2. Impulse response functions to a capital depreciation shock

NOTES. All the IRFs, except employment and effort, are expressed relative to trend. Parameters are calibrated as reported in section 4.1.



Figure 3. Labour input and wage responses

NOTES. Parameters are calibrated as in section 4.1 Each point in the graph is the combination of the real wage deviation from trend in the fourth quarter after the shock and effort (panel A) and employment (panel B) deviation in the same quarter.



Figure 4. Jobless recovery. Impulse response functions.

NOTES. Parameters are calibrated as in section 4.1 and d=0.85 consistent with a steady state wage markup of 12%. All the IRFs, except employment and effort, are expressed relative to trend.



Figure 5. Jobless recovery. Dynamic simulation of employment and undetrended GDP and labour productivity in levels.

NOTES. The figures show the dynamic simulation of N_t and undetrended Y_t and $\alpha(Y_t/N_t)$. Parameters are calibrated as in section 4.1. and d = 0.85. Light blue shaded area indicates jobless recovery: Y_t and $\alpha(Y_t/N_t)$ grow at around their balanced growth rate γ while N_t stays persistently at the depressed level.

Figure 6. The detection probability and the discipline device role of unemployment



NOTES. As shown in Appendix C, in steady state $\mathcal{I} = \beta \frac{(1-d)}{[1+\delta\beta(1-x)]} \frac{[1+\delta(1-x)]d}{[1-(1-\delta-d)(1-x)]} \frac{\alpha \varkappa}{\Theta} \frac{[\delta+x(1-\delta)]}{x}$. Parameters are calibrated as reported in section 4.1.

Figure 7. Impulse response functions to a capital depreciation shock when effort is observable



NOTES. Parameters are calibrated as reported in section 4.1. Effort is perfectly observable. All the IRFs, except employment and effort, are expressed relative to trend.

Appendix A

The model is consistent with balanced growth: trend productivity growth determines trend increases in output, consumption, wages, capital and investment, without affecting employment and workers' effort. Given the presence of a trend, before log-linearizing the model, I derive its stationary representation. In what follows the "~" superscript indicates that level variables are expressed in terms of stationary ratios (i.e. given the generic variable Z_t , we have $\tilde{Z}_t = \frac{Z_t}{\gamma^t}$). The model is then log-linearized around the steady state of the scaled (stationary) variables. Lower case letters with a "~" thus denote logdeviations of the corresponding detrended variables from the stationarized steady state.

Steady state relations

Using the *no-shirking condition* (20) I define:

$$\Psi \equiv \underbrace{\frac{\widetilde{C}}{\widetilde{W}} \frac{\mathcal{E}}{d}}_{MRS_{C,N}/d\widetilde{W}} = \frac{\left[1 + \beta \left(1 - x\right)\delta\right]}{\left[1 - \beta \left(1 - x\right)\left(1 - \delta - d\right)\right]}$$
(1A)

Equation (22) implies that:

$$\Omega \equiv \underbrace{\frac{\alpha Y}{NW}}_{\widetilde{MPN}/\widetilde{W}} = 1 + h_w \left[1 - (1 - \delta) \beta \right]$$
(2A)

where I define $h_w \equiv \frac{Bx^{\eta}}{\widetilde{W}}$. The steady state investment to capital ratio is:

$$\frac{\widetilde{I}}{\widetilde{K}} = \gamma - 1 + \overline{\varrho} \tag{3A}$$

where $\overline{\varrho}$ is the mean depreciation rate. Equation (6) implies $R^k = \frac{\gamma}{\beta} - 1 + \overline{\varrho}$, thus taking into account that $R^k = (1 - \alpha) \frac{\tilde{Y}}{\tilde{K}}$, we have:

$$\Theta_I \equiv \frac{\widetilde{I}}{\widetilde{Y}} = \frac{(1-\alpha)\left(\gamma - 1 + \overline{\varrho}\right)}{\frac{\gamma}{\beta} - 1 + \overline{\varrho}}$$
(4A)

which implies:

$$\Theta_C \equiv \frac{\widetilde{C}}{\widetilde{Y}} = 1 - \Theta_I - \delta h_w \frac{\alpha}{\Omega}$$
(5A)

Note that hiring costs represent a fraction $h_y \equiv \frac{Bx^{\eta}H}{\tilde{Y}^r} = h_w \frac{\alpha\delta}{\Omega}$ of GDP.

The complete loglinear model

Leaving out the stochastic process, the complete system of log-linear equations in stationary form is as follows:

Technology

$$\widetilde{y}_t = \alpha n_t + \alpha \varkappa \varepsilon_t + (1 - \alpha) \widetilde{k}_t$$

Job Finding Rate
 $\delta x_t = n_t - (1 - \delta)(1 - x)n_{t-1}$

Aggregate Resource constraint

 $\widetilde{y}_t = \Theta_C \widetilde{c}_t + \frac{\alpha h_w}{\Omega} \left[n_t - (1 - \delta) n_{t-1} + \delta \eta x_t \right] + \Theta_I \widetilde{i}_t$

Rental Cost
$$r_t^k = \widetilde{y}_t - \widetilde{k}_t$$

Capital Accumulation $\widetilde{k}_{t+1} = \frac{(1-\overline{\varrho})}{\gamma} \widetilde{k}_t - \frac{\varrho}{\gamma} \varrho_t + \frac{(\gamma-1+\overline{\varrho})}{\gamma} \widetilde{i}_t$

> Investment and Capital $i_t - k_t = \mu q_t^{Tobin}$

Tobin Q $q_t^{Tobin} = \beta \mathbb{E}_t q_{t+1}^{Tobin} + (c_t - \mathbb{E}_t c_{t+1}) + \left[1 - \frac{\beta}{\gamma} \left(1 - \overline{\varrho} \right) \right] \mathbb{E}_t r_{t+1}^k - \frac{\beta}{\gamma} \overline{\varrho} \varrho_t$

Labour Demand

 $\widetilde{w}_t = \Omega \left(\widetilde{y}_t - n_t \right) - \eta h_w x_t + (1 - \delta) \beta h_w \mathbb{E}_t \left\{ \eta x_{t+1} + \widetilde{c}_t - \widetilde{c}_{t+1} \right\}$

Effort/Employment Relation

$$n_{t} = (1 - \delta - d) (1 - x) n_{t-1} - x (1 - \delta - d) x_{t} + \frac{\varkappa \Omega}{\Psi} [\widetilde{y}_{t} - \widetilde{c}_{t} - \varepsilon_{t}] + \frac{\varkappa \Omega}{\Psi} \delta (1 - x) [\widetilde{y}_{t-1} - \widetilde{c}_{t-1} - \varepsilon_{t-1}] - \delta x \frac{\varkappa \Omega}{\Psi} x_{t}$$

No-Shirking Condition

 $\widetilde{w}_{t} = \Psi \left[\widetilde{c}_{t} + \varepsilon_{t} \right] - \beta \left(1 - x \right) \left[\delta + \left(1 - \delta - d \right) \Psi \right] \mathbb{E}_{t} \left[\widetilde{c}_{t} - \widetilde{c}_{t+1} \right] +$ $+\beta x \left[\delta + (1 - \delta - d)\Psi\right] \mathbb{E}_{t} x_{t+1} - \beta (1 - x) \delta \mathbb{E}_{t} \widetilde{w}_{t+1} - \beta (1 - x) (1 - \delta - d)\Psi \mathbb{E}_{t} \left[\widetilde{c}_{t+1} + \varepsilon_{t+1}\right]$

Appendix B

As long as the real wage is consistent with a non-negative surplus for both firms and workers, workers and firms will not volontarily put an end to their relationship, over the life of their relationship. The bargaining set is defined by the range of wage levels above the workers' reservation wage W_t^L and below the firm's reservation wage W_t^U : $W_t \in [W_t^L, W_t^U]$. Subtracting (16) from (12) yields:

$$S_t^H = W_t - C_t \mathcal{E}_t + (1 - \delta) \,\beta \mathbb{E}_t \frac{C_t}{C_{t+1}} \left(1 - X_{t+1}\right) S_{t+1}^H \tag{6A}$$

The worker's reservation wage is the lowest wage such that $S_t^H \ge 0$ and can be expressed as follows:

$$W_t^L = C_t \mathcal{E}_t - (1 - \delta) \beta \mathbb{E}_t \frac{C_t}{C_{t+1}} (1 - X_{t+1}) S_{t+1}^H$$
(7A)

The surplus from an existing employment relation accruing to firm is given by:

$$S_{j,t}^F = \alpha Y_{j,t} / N_{j,t} - W_{j,t} + (1 - \delta) \mathbb{E}_t Q_{t,t+1} S_{j,t+1}^F$$
(8A)

The corresponding reservation wage for the firm is the highest wage such that $S_{jt}^F \ge 0$:

$$W_{j,t}^{U} = \alpha Y_{j,t} / N_{j,t} + (1 - \delta) \mathbb{E}_{t} Q_{t,t+1} S_{j,t+1}^{F}$$
(9A)

Under the continuing assumption that the firm is maximizing profits, it follows from (8A) and (22) that $S_{j,t}^F = G_t \ \forall j$.

In order to show that, under the calibration discussed in section 4.1., wages have a negligible probability of falling outside the bargaining set, I first generate artificial time series of 12000 observations from the model²⁸ and then compute a time series for the reservation wage of workers, the reservation wage of firms and the equilibrium wage, under different calibrated values of the probability of being caught shirking. The first panel in Figure 8 plots the three time series when d = 0.25, a value which implies a steady state wage markup of 213%; the second panel reports the three time series when d = 0.95, a value which implies a steady state wage markup of 3.74%. In both cases, the real wage stays in the bargaining set, moving close to the upper bound when d = 0.25, while lying close to the lower bound when d = 0.95.

²⁸In order to preserve the clarity of the figure, I plot only 1000 observations.



Figure 8. Efficiency wages and the bargaining set

NOTES. Parameters are calibrated as in section 4.1

Appendix C

This appendix sheds light on the dependence of the discipline device role of unemployment (as quantified in 27), evaluated in steady state, on the detection probability d.

Combining (20) with (17) one gets the steady state surplus accruing to the household from an existing employment relation:

$$S^{H} = \frac{1-d}{\left[1+\delta\beta\left(1-x\right)\right]} \frac{C\mathcal{E}}{d}$$
(10A)

Accordingly, one has:

$$\mathcal{I} = \beta \frac{(1-d)}{[1+\delta\beta (1-x)]} \mathcal{E}$$
(11A)

Equation (23) evaluated in steady state yields the steady state level of effort \mathcal{E} . The latter is zero when d = 0 and is monotonically strictly increasing with d:

$$\mathcal{E} = \frac{d}{x} \frac{\alpha \varkappa [\delta + (1 - \delta) x]}{\Theta_C} \frac{[1 + \delta (1 - x)]}{[1 - (1 - \delta - d) (1 - x)]}$$
(12A)

Finally, the previous equation (12A) can be combined with (11A) to get:

$$\mathcal{I} = d\left(1-d\right)\beta \frac{\alpha \varkappa \left[\delta + (1-\delta)x\right]}{\Theta_C \left[1 - (1-\delta-d)\left(1-x\right)\right]} \frac{\left[1+\delta\left(1-x\right)\right]}{\left[1+\delta\beta\left(1-x\right)\right]x}$$
(13A)

Note that in steady state unemployment plays no role as a method of discipline, i.e. $\mathcal{I} = 0$, both when the detection probability is zero and when it is equal to one. The dependence of the incentive discipline device role of unemployment evaluated in steady state on the detection probability d is illustrated in Figure 6, that shows that when the detection probability is relatively small, increases in d strengthen the role of unemployment as a method of discipline and, vice-versa, weaken it when d is relatively high.

References

 Alexopoulos, Michelle (2004), "Unemployment and the Business Cycle", Journal of Monetary Economics 51, 277–298

- [2] Andolfatto, David and Glenn MacDonald (2004), "Jobless Recoveries" EconWPA, Macroeconomics, 0412014.
- [3] Atkinson, Anthony and Thomas Piketty (2006), "Top incomes over the twentieth century." Oxford: Oxford University Press, 2006.
- [4] Baker, George (1992), "Incentive contracts and performance measurement", *Journal* of Political Economy 100, 598 614.
- [5] Barnichon, Regis (2011), "Vacancy posting, Job separation and Unemployment Fluctuations," *Journal of Economic Dynamics and Control*, Forthcoming.
- [6] Barro, Robert J. (2006), "Rare disasters and asset markets in the twenthieth century," *Quarterly Journal of Economics* 99, 823-866.
- [7] Bernanke, Ben S. (2003), "The Jobless Recovery", Remarks by Governor At the Global Economic and Investment Outlook Conference, Carnegie Mellon University, Pittsburgh, Pennsylvania November 6, 2003 www.federalreserve.gov/ boarddocs /speeches / 2003/ 200311062/ default.htm
- [8] Blanchard Olivier J. and Peter Diamond (1990), "The Cyclical Behavior of the Gross Flows of US Workers," *Brookings Papers on Economic Activity*, 1, 1-60.
- [9] Blanchard Olivier J. and Jordi Gali (2010), "Labour Market Frictions and Monetary Policy: A New Keynesian Model with Unemployment," *American Economic Journal: Macroeconomics*, 2(2), 1-30.
- [10] Champagne, Julien and André Kurmann, (2011), "The Great Increase in Relative Volatility of Real Wages in the United States", mimeo.
- [11] Chen, Jinzhu, Prakash Kannan, Prakash Loungani and Bharat Trehan (2011), "New Evidence on Cyclical and Structural Sources of Unemployment", IMF Working Paper WP/11/106.
- [12] Danthine and Donaldson (1995), "Non Walrasian Economies", in Thomas F. Cogley, ed., Frontiers of Business Cycle research, Princepton University Press (1995) 217-242.
- [13] Elsby, Michael, Bart Hobijn, and Aysegul Sahin, forthcoming, "The labour Market in the Great Recession." *Brookings Papers on Economic Activity.*

- [14] Faia, Ester (2009), "Ramsey monetary policy with labor market frictions," *Journal* of Monetary Economics 56 570-581.
- [15] Fischer, Jonas D.M. (2006), "The Dynamic Effects of Neutral and Investment-Specific Technology Shocks," *Journal of Political Economy*, 114, 413-451.
- [16] Furlanetto, Francesco and Martin Seneca (2011), "New Perspective on Depreciation Shocks as a Source of Business Cycle Fluctuations," Norges Bank WP 2011/02.
- [17] Gali, Jordi, David Lopez-Salido and Javier Vallés (2007), "Understanding the Effects of Government Spending on Consumption," *Journal of the European Economic Association*, 5(1), 227-270.
- [18] Gali, Jordi, Frank Smets and Rafael Wouters (2011), "Slow Recoveries: A Structural Interpretation", mimeo Universitat Pompeu Fabra.
- [19] Gertler, Mark, Luca Sala and Antonella Trigari (2008), "An Estimated Monetary DSGE Model with Unemployment and Staggered Nominal Wage Bargaining," *Jour*nal of Money Credit and Banking, 40 (8): 1713-1764.
- [20] Gertler, Mark and Peter Karadi (2011), "A Model of Unconventional Monetary Policy," Journal of Monetary Economics, 58, 17-34.
- [21] Gertler, Mark and Nobuhiro Kiyotaki, (2011), "Financial intermediation and credit policy in business cycle analysis," in B. Friedman and M. Woodford (ed.), Handbook of Montary Economics, ch. 11.
- [22] Gomme, Paul (1999), "Shirking, Unemployment and Aggregate Fluctuations", International Economic Review, 40(1) 3-21.
- [23] Gordon, Robert J. (1979), "The "End-of-Expansion" Phenomenon in Short-Run Productivity Behavior," Brookings Papers on Economic Activity, vol. 10(2), pages 447-462.
- [24] Gordon, Robert J. (1993), "The Jobless Recovery: Does it Signal a New Era of Productivity-led Growth?" Brookings Papers on Economic Activity, 1:1993, 217-306.
- [25] Gordon, Robert J. (2003), "Exploding Productivity Growth: Context, Causes, and Implications," Brookings Papers on Economic Activity, 2, 207-79.

- [26] Gordon, Robert J. (2010), "The Demise of Okun's Law and of Procyclical Fluctuations in Conventional and Unconventional Measures of Productivity" mimeo Northwestern University
- [27] Groshen, Erica, and Simon Potter (2003), "Has Structural Change Contributed to a Jobless Recovery?" Federal Reserve Bank of New York, Current Issues in Economics and Finance, 9(8).
- [28] Hall, Robert E. (2005), "Employment Efficiency and Sticky Wages: Evidence from Flows in the Labor Market," *The Review of Economics and Statistics*, 87(3), 397-407.
- [29] Hall, Robert E. (2010), "Why Is Unemployment Still So High Two Years after the Financial Crisis", mimeo, Stanford University.
- [30] Justiniamo, Alejandro, Giorgio E. Primiceri (2008), "The Time Varying Volatility of Macroeconomic Fluctuations," *American Economic Review*, 98, 604-641.
- [31] Justiniano, Alejandro, Giorgio Primiceri G. and Andrea Tambalotti (2010), "Investment shocks and business cycles," *Journal of Monetary Economics* 57, 132-145.
- [32] Kimball, Miles S. (1994), "Labour Market Dynamics when Unemployment is a Worker Discipline Device," *The American Economic Review*, 84(4), 1045-1059.
- [33] King, Robert and Mark Watson (1996), "Money, Prices, Interest Rates and the Business Cycle," *Review of Economics and Statistics*, 78: 35-53.
- [34] Lazear Edward P (2000), "Performance Pay and Productivity" American Economic Review, vol. 90 no. pp. 1346-1361.
- [35] Lemieux, Thomas, W. Bentley MacLeod and Daniel Parent (2009), "Performance Pay and Wage Inequality", *Quarterly Journal of Economics*, 124(1), 1-49.
- [36] Lilien, David M. (1982), "Sectoral Shifts and Cyclical Unemployment", Journal of Political Economy, vol. 90, no. 4, 777-93
- [37] Liu, Z., D.F. Waggoner and T. Zha (2010), Sources of Macroeconomic Fluctuations: A Regime-Switching DSGE Approach, FRB San Francisco WP, 2009-01.
- [38] Loungani, Prakash and Bharat Trehan (1997), "Explaining unemployment: sectoral vs aggregate shocks," *Economic Review*, Federal Reserve Bank of San Francisco, 3-15.

- [39] Monacelli, Tommaso, Vincenzo Quadrini and Antonella Trigari (2011), "Financial Markets and Unemployment", NBER Working Paper No. w17389.
- [40] Nucci, Francesco and Marianna Riggi (2011), "Performance pay and shifts in macroeconomic correlations", Temi di discussione 800, Bank of Italy Working Paper.
- [41] Ohanian, Lee E. (2010a) "The Economic Crisis from a Neoclassical Perspective", Journal of Economic Perspective, 24(4), 45-66.
- [42] Ohanian, Lee E. (2010b) "Understanding Economic Crises: The Great Depression and the 2008 Recession," *The Economic Record*, 86, 2-6.
- [43] Ohanian, Lee E. (forthcoming) "What or Who Started the Great Depression?", forthcoming, *Journal of Economic Theory*
- [44] Oliner, Stephen D., Daniel E. Sichel and Kevin J. Stiroh (2007), "Explaining a Productivity Decade," Brookings Papers on Economic Activity, 1, 81-137.
- [45] Piketty, Thomas and Emmanuel Saez (2006), "The Evolution of Top Incomes: A Historical and International Perspective," American Economic Review 96(2): 200-205.
- [46] Shapiro, Carl and Joseph E. Stiglitz (1984), "Equilibrium Unemployment as a Worker Discipline Device", American Economic Review, 74(3) 433-444.
- [47] Shimer, Robert (2005), "The Cyclical Behavior of Equilibrium Unemployment and Vacancies," *American Economic Review*, 95(1): 25-49.
- [48] Shimer, Robert (2007), "Reassessing the Ins and Outs of Unemployment," NBER WP 13421.
- [49] Shimer, Robert (2010), "Wage Rigidities and Jobless Recoveries" mimeo University of Chicago
- [50] Solow, Robert (1979), "Another possible source of wage stickiness" Journal of Macroeconomics, 1(1), 79-82.
- [51] Uhlig, Harald and Yexiao Xu (1996), "Effort and the Cycle: Cyclical Implications of Efficiency Wages", Tilburg University, Discussion Paper 1996-49
- [52] Van Rens, Thijs (2005), "Organizational Capital and Employment Fluctuations", Universitat Pompeu Fabra, mimeo.

- [53] Woodford, Michael (1994), "Notes on Dynamic Efficiency Wage Models", mimeo.
- [54] Yellen, Janet L. (2010), "The Outlook for the Economy and Monetary Policy", President's Speech University of San Diego San Diego http://www.frbsf.org/news/speeches/2010/janet_yellen0222.html

RECENTLY PUBLISHED "TEMI" (*)

- N. 848 Bank balance sheets and the transmission of financial shocks to borrowers: evidence from the 2007-2008 crisis, by Emilia Bonaccorsi di Patti and Enrico Sette (January 2012).
- N. 849 *The role of financial investments in agricultural commodity derivatives markets*, by Alessandro Borin and Virginia Di Nino (January 2012).
- N. 850 Mapping local productivity advantages in Italy: industrial districts, cities or both?, by Valter Di Giacinto, Matteo Gomellini, Giacinto Micucci and Marcello Pagnini (January 2012).
- N. 851 The impact of monetary policy shocks on commodity prices, by Alessio Anzuini, Marco J. Lombardi and Patrizio Pagano (February 2012).
- N. 852 *Statistical models for measuring job satisfaction*, by Romina Gambacorta and Maria Iannario (February 2012).
- N. 853 Forecasting world output: the rising importance of emerging economies, by Alessandro Borin, Riccardo Cristadoro, Roberto Golinelli and Giuseppe Parigi (February 2012).
- N. 854 Entry dynamics as a solution to the puzzling behaviour of real marginal costs in the Ghironi-Melitz model, by Alberto Felettigh (February 2012).
- N. 855 Trade openness and international fragmentation of production in the European Union: the new divide?, by Paolo Guerrieri and Filippo Vergara Caffarelli (February 2012).
- N. 856 Optimal dynamic public communication, by Marcello Miccoli (February 2012).
- N. 857 Modelling loans to non-financial corporations in the euro area, by Christoffer Kok Sørensen, David Marqués Ibáñez and Carlotta Rossi (February 2012).
- N. 858 Economic (in)stability under monetary targeting, by Luca Sessa (March 2012).
- N. 859 *Optimal fiscal policy when agents fear government default*, by Francesco Caprioli, Pietro Rizza and Pietro Tommasino (March 2012).
- N. 860 The effects of unemployment benefits in Italy: evidence from an institutional change, by Alfonso Rosolia and Paolo Sestito (March 2012).
- N. 861 *Monetary policy and the flow of funds in the euro area*, by Riccardo Bonci (March 2012).
- N. 862 Does gender matter for public spending? Empirical evidence from Italian municipalities, by Massimiliano Rigon and Giulia M. Tanzi (April 2012).
- N. 863 House price cycles in emerging economies, by Alessio Ciarlone (April 2012).
- N. 864 Measuring the underground economy with the currency demand approach: a reinterpretation of the methodology, with an application to Italy, by Guerino Ardizzi, Carmelo Petraglia, Massimiliano Piacenza and Gilberto Turati (April 2012).
- N. 865 *Corporate geography in multi-plant firms*, by Rita Cappariello, Stefano Federico and Roberta Zizza (April 2012).
- N. 866 *Don't stand so close to me: the urban impact of immigration*, by Antonio Accetturo, Francesco Manaresi, Sauro Mocetti and Elisabetta Olivieri (April 2012).
- N. 867 Disinflation effects in a medium-scale New Keynesian model: money supply rule versus interest rate rule, by Guido Ascari and Tiziano Ropele (April 2012).
- N. 868 *The economic costs of organized crime: evidence from southern Italy*, by Paolo Pinotti (April 2012).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- F. PANETTA, F. SCHIVARDI and M. SHUM, *Do mergers improve information? Evidence from the loan market*, Journal of Money, Credit, and Banking, v. 41, 4, pp. 673-709, **TD No. 521 (October 2004).**
- M. BUGAMELLI and F. PATERNÒ, *Do workers' remittances reduce the probability of current account reversals?*, World Development, v. 37, 12, pp. 1821-1838, **TD No. 573 (January 2006).**
- P. PAGANO and M. PISANI, *Risk-adjusted forecasts of oil prices*, The B.E. Journal of Macroeconomics, v. 9, 1, Article 24, **TD No. 585 (March 2006).**
- M. PERICOLI and M. SBRACIA, The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems, International Finance, v. 12, 2, pp. 123-150, TD No. 586 (March 2006).
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, v. 39, 2, pp.187-199, **TD No. 597 (September 2006).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, v. 5, 4, pp. 393-409, **TD No. 601 (September 2006).**
- F. BALASSONE, D. FRANCO and S. ZOTTERI, *The reliability of EMU fiscal indicators: risks and safeguards*, in M. Larch and J. Nogueira Martins (eds.), Fiscal Policy Making in the European Union: an Assessment of Current Practice and Challenges, London, Routledge, **TD No. 633 (June 2007).**
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, v. 19, 2, pp. 222-239, **TD No. 637 (June 2007)**.
- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- V. DI GIACINTO and G. MICUCCI, The producer service sector in Italy: long-term growth and its local determinants, Spatial Economic Analysis, Vol. 4, No. 4, pp. 391-425, TD No. 643 (September 2007).
- F. LORENZO, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: estimates for the euro area*, Journal of Public Economics, v. 93, 3-4, pp. 559-585, **TD No. 652** (November 2007).
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, v. 53, 8, pp. 996-1009, **TD No. 653** (November 2007).
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers, Giornale degli Economisti e Annali di Economia, v. 68, 1, pp. 25-52, TD No. 655 (January 2008).
- F. MACCHERONI, M. MARINACCI, A. RUSTICHINI and M. TABOGA, *Portfolio selection with monotone mean*variance preferences, Mathematical Finance, v. 19, 3, pp. 487-521, **TD No. 664 (April 2008).**
- M. AFFINITO and M. PIAZZA, What are borders made of? An analysis of barriers to European banking integration, in P. Alessandrini, M. Fratianni and A. Zazzaro (eds.): The Changing Geography of Banking and Finance, Dordrecht Heidelberg London New York, Springer, TD No. 666 (April 2008).
- A. BRANDOLINI, On applying synthetic indices of multidimensional well-being: health and income inequalities in France, Germany, Italy, and the United Kingdom, in R. Gotoh and P. Dumouchel (eds.), Against Injustice. The New Economics of Amartya Sen, Cambridge, Cambridge University Press, TD No. 668 (April 2008).
- G. FERRERO and A. NOBILI, *Futures contract rates as monetary policy forecasts*, International Journal of Central Banking, v. 5, 2, pp. 109-145, **TD No. 681 (June 2008).**
- P. CASADIO, M. LO CONTE and A. NERI, Balancing work and family in Italy: the new mothers' employment decisions around childbearing, in T. Addabbo and G. Solinas (eds.), Non-Standard Employment and Qualità of Work, Physica-Verlag. A Sprinter Company, TD No. 684 (August 2008).
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**

- G. ASCARI and T. ROPELE, *Trend inflation, taylor principle, and indeterminacy*, Journal of Money, Credit and Banking, v. 41, 8, pp. 1557-1584, **TD No. 708** (May 2007).
- S. COLAROSSI and A. ZAGHINI, *Gradualism, transparency and the improved operational framework: a look at overnight volatility transmission*, International Finance, v. 12, 2, pp. 151-170, **TD No. 710 (May 2009).**
- M. BUGAMELLI, F. SCHIVARDI and R. ZIZZA, *The euro and firm restructuring*, in A. Alesina e F. Giavazzi (eds): Europe and the Euro, Chicago, University of Chicago Press, **TD No. 716 (June 2009).**
- B. HALL, F. LOTTI and J. MAIRESSE, *Innovation and productivity in SMEs: empirical evidence for Italy*, Small Business Economics, v. 33, 1, pp. 13-33, **TD No. 718 (June 2009).**

- A. PRATI and M. SBRACIA, Uncertainty and currency crises: evidence from survey data, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, v. 42, 19-21, pp. 2399-2415, **TD No. 458 (December 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, v. 50, 4, pp. 833-857, **TD No. 591 (May 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area*, International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, *New Eurocoin: Tracking Economic Growth in Real Time*, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, **TD No. 631 (June 2007).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, Cyclical asymmetry in fiscal variables in the EU, Empirica, TD No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, O. GIANMARCO I.P. and P. GIOVANNI, The labor market impact of immigration on the western german labor market in the 1990s, European Economic Review, v. 54, 4, pp. 550-570, TD No. 687 (August 2008).
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums,* Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, Dropping the Books and Working Off the Books, Labour, v. 24, 2, pp. 139-162, TD No. 702 (January 2009).
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, v. 14, 5, pp. 677-708, **TD No. 706** (March 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740 (January 2010).**
- M. AFFINITO and E. TAGLIAFERRI, *Why do (or did?) banks securitize their loans? Evidence from Italy*, Journal of Financial Stability, v. 6, 4, pp. 189-202, **TD No. 741 (January 2010).**

- S. FEDERICO, Outsourcing versus integration at home or abroad and firm heterogeneity, Empirica, v. 37, 1, pp. 47-63, **TD No. 742** (February 2010).
- V. DI GIACINTO, *On vector autoregressive modeling in space and time*, Journal of Geographical Systems, v. 12, 2, pp. 125-154, **TD No. 746 (February 2010).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, v. 34, 9, pp. 1791-1812, **TD No. 747** (March 2010).
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- P. CIPOLLONE, P. MONTANARO and P. SESTITO, Value-added measures in Italian high schools: problems and findings, Giornale degli economisti e annali di economia, v. 69, 2, pp. 81-114, TD No. 754 (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, *Asset-based measurement of poverty*, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757** (April 2010).
- S. DI ADDARIO and D. VURI, Entrepreneurship and market size. the case of young college graduates in Italy, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**
- A. CALZA and A. ZAGHINI, *Sectoral money demand and the great disinflation in the US*, Journal of Money, Credit, and Banking, v. 42, 8, pp. 1663-1678, **TD No. 785 (January 2011).**

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximun entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, **TD No. 641** (September 2007).
- E. CIAPANNA, Directed matching with endogenous markov probability: clients or competitors?, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, *Local and global agglomeration patterns: two econometrics-based indicators*, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- R. GIORDANO and P. TOMMASINO, What determines debt intolerance? The role of political and monetary *institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union: the Case of Italy*, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., **TD No. 749 (March 2010).**
- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**

- G. BARBIERI, C. ROSSETTI and P. SESTITO, The determinants of teacher mobility: evidence using Italian teachers' transfer applications, Economics of Education Review, v. 30, 6, pp. 1430-1444, TD No. 761 (June 2010).
- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, **TD No.** 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- A. FELETTIGH and S. FEDERICO, *Measuring the price elasticity of import demand in the destination markets of italian exports*, Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**
- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807** (April 2011).
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837** (November 2011).
- F. CAPRIOLI, P. RIZZA and P. TOMMASINO, *Optimal fiscal policy when agents fear government default,* Revue Economique, v. 62, 6, pp. 1031-1043, **TD No. 859 (March 2012).**

- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate*, International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010).**
- A. ACCETTURO and G. DE BLASIO, *Policies for local development: an evaluation of Italy's "Patti Territoriali"*, Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, **TD No. 789 (January 2011).**

FORTHCOMING

- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No.** 578 (February 2006).
- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, **TD No. 600 (September 2006).**
- M. BRATTI, D. CHECCHI and G. DE BLASIO, *Does the expansion of higher education increase the equality of educational opportunities? evidence from Italy*, Labour, **TD No. 679 (June 2008).**
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, **TD No. 691 (September 2008).**
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, **TD No. 696 (December 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, **TD No. 698 (December 2008).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis*, Journal of European Economic Association, **TD No. 704** (March 2009).

- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, **TD No. 709 (May 2009).**
- Y. ALTUNBAS, L. GAMBACORTA, and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, **TD No. 712 (May 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian selection*, Journal of International Economics, **TD No. 728 (October 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, **TD No. 732 (November 2009).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, **TD No. 835 (February 2011).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets*, International Journal of Industrial Organization, **TD No. 760 (June 2010).**
- L. MONTEFORTE and G. MORETTI, *Real-Time forecasts of inflation: the role of financial variables*, Journal of Forecasting, **TD No. 767 (July 2010).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, **TD No. 802 (March 2011).**
- M. AFFINITO, Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy, Journal of Banking and Finance, **TD No. 826 (October 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Households Savings in China*, Chinese Economic and Business Studies, **TD No. 838 (November 2011).**